Author’s response to reviews

Title: Educational difference in the prevalence of osteoporosis in postmenopausal women: a descriptive study in northern Iran

Authors:

Mohsen Maddah (Maddahm@yahoo.com)
Hajar S Sharami (SharamiH@yahoo.com)
Majid Karandish (karandish@yaho.com)

Version: 4 Date: 19 September 2011

Author’s response to reviews: see over
Title: Educational difference in the prevalence of osteoporosis in Post-menopausal women: a descriptive study in northern Iran

Dear Editor,

We thank you and the reviewer for his helpful comments. We applied the requested changes in the revised manuscript as follow:

In table we added p-values in the last two columns.

We changed "...among Iranian population, in all age groups, peak bone mass is lower than European or American population" to "...is lower than in European or American populations".

We changed "Education is one of most commonly used measure of socioeconomic status (SES) in epidemiological studies" to "...is one of the more commonly used measures..."

We change… low social level should be carefully evaluated for signs of osteoporosis during routine physical examinations." to "...from lower social classes in this community. Therefore, women from lower social classes...".

Best regards,

Mohsen Maddah

Seyed Hajar Sharami

Majid Karandish